A randomised controlled study of the effectiveness of breathing retraining exercises taught by a physiotherapist either by instructional DVD or in face-to-face sessions in the management of asthma in adults

Mike Thomas, 1* Anne Bruton, 2 Paul Little, 1
Stephen Holgate, 1 Amanda Lee, 3 Lucy Yardley, 4
Steve George, 1 James Raftery, 1 Jennifer Versnel, 5
David Price, 3 Ian Pavord, 6 Ratko Djukanovic, 1
Michael Moore, 1 Sarah Kirby, 4 Guiqing Yao, 1
Shihua Zhu, 1 Emily Arden-Close, 7
Manimekalai Thiruvothiyur, 3 Frances Webley, 8
Mark Stafford-Watson, 9 Elizabeth Dixon 8
and Lynda Taylor 8

¹Faculty of Medicine, University of Southampton, Southampton, UK
²Faculty of Health Sciences, University of Southampton, Southampton, UK
³Institute of Applied Health Sciences, University of Aberdeen, Aberdeen, UK
⁴School of Psychology, University of Southampton, Southampton, UK
⁵Asthma UK, London, UK
⁶Nuffield Department of Medicine, University of Oxford, Oxford, UK
⁷Department of Psychology, Bournemouth University, Bournemouth, UK
⁸Southampton Clinical Trials Unit, University of Southampton, Southampton, UK

⁹Patient and public involvement representative

Declared competing interests of authors: Mike Thomas is a member of the Health Technology Assessment (HTA) Primary Care, Community and Preventive Interventions (PCCPI) Panel. In the last 3 years he has received speaker's honoraria for speaking at sponsored meetings or satellite symposia at conferences from the following companies, marketing respiratory and allergy products: Aerocrine, GlaxoSmithKline (GSK) and Novartis International AG. He has received honoraria for attending advisory panels with Aerocrine, AstraZeneca, Boehringer Ingelheim, GSK and Novartis. He has received sponsorship to attend international scientific meetings from GSK and AstraZeneca and has received funding for research projects from GSK. He is a member of the British Thoracic Society (BTS)/Scottish Intercollegiate Guidelines Network (SIGN) Asthma Guideline Group and the National Institute for Health and Care Excellence (NICE) Asthma Guideline Group. Paul Little is Editor-in-Chief of the *Programme Grants for Applied Research* journal and is a member of the National Institute for Health Research (NIHR) Journals

^{*}Corresponding author D.M.Thomas@soton.ac.uk

Library Board. Lucy Yardley Is a member of the Public Health Research journal Research Funding Board and a member of the HTA Efficient Study Designs Board and reports grants from the NIHR during the conduct of the study. James Raftery is a member of the NIHR Journals Library Editorial Group and the HTA and EME Editorial Board and was previously Director of the Wessex Institute and head of the NIHR Evaluation, Trials and Studies Coordinating Centre (NETSCC). He is also a former member of the NIHR HSDR Research Led Board. Ian Pavord has received speaker's honoraria for speaking at sponsored meetings in the last 5 years from AstraZeneca, Boehringer Ingelheim, Aerocrine, Almirall Ltd, Novartis and GSK. He has received honoraria for attending advisory panels with Almirall, AstraZeneca, Boehringer Ingelheim, GSK, MSD, Schering-Plough, Novartis, Dey Pharma, Napp Pharmaceuticals and RespiVert Ltd. He has received sponsorship to attend international scientific meetings from Boehringer Ingelheim, GSK, AstraZeneca and Napp. He is Chief Medical Advisor to Asthma UK, a member of the UK Department of Health Asthma Strategy Group, a member of the BTS/SIGN Asthma Guideline Group and joint Editor-in-Chief of Thorax. Neither Ian Pavord nor any member of his family has any shares in pharmaceutical companies. David Price reports other Board Membership (fees paid to Research in Real Life Ltd) from Aerocrine, Almirall, Amgen Inc., AstraZeneca, Boehringer Ingelheim, Chiesi Ltd, Meda, Mundipharma, Napp, Novartis and Teva Pharmaceutical Industries Ltd; consultancy (fees paid to Research in Real Life Ltd) from Almirall, Amgen, AstraZeneca, Boehringer Ingelheim, Chiesi, GSK, Meda, Mundipharma, Napp, Novartis, Pfizer Inc. and Teva; grants from the UK NHS, British Lung Foundation, Aerocrine, AKL Ltd, Almirall, AstraZeneca, Boehringer Ingelheim, Chiesi, Eli Lilly and Co., GSK, Meda, Merck & Co., Inc., Mundipharma, Napp, Novartis, Orion, Pfizer, Respiratory Effectiveness Group, Takeda, Teva and Zentiva; lectures/speaking engagement fees (paid to Research in Real Life Ltd) from Almirall, AstraZeneca, Boehringer Ingelheim, Chiesi, Cipla Ltd, GSK, Kyorin Pharmaceutical Co., Inc., Meda, Merck, Mundipharma, Novartis, Pfizer, Skyepharma, Takeda and Teva; manuscript preparation fees (paid to Research in Real Life Ltd) from Mundipharma and Teva; payment for travel/accommodation/meeting expenses (paid to Research in Real Life Ltd) from Aerocrine, Boehringer Ingelheim, Mundipharma, Napp, Novartis and Teva; funding for patient enrolment or completion of research (paid to Research in Real Life Ltd) from Almirall, Chiesi, Teva and Zentiva; and payment for the development of educational materials (paid to Research in Real Life Ltd) from GSK and Novartis, outside the submitted work. In addition, David Price has an AKL Ltd patent pending and owns shares in AKL Ltd, which produces phytopharmaceuticals. He owns 80% of Research in Real Life Ltd (which is subcontracted by Observational and Pragmatic Research Institute Pte Ltd), 75% of the social enterprise Optimum Patient Care Ltd and 75% of the Observational and Pragmatic Research Institute Pte Ltd. Ratko Djukanovic has received fees for lectures at symposia organised by Novartis and Teva and for consultation for these two companies as a member of advisory boards. He is a co-founder and current consultant, and has shares in, Synairgen, a University of Southampton spin-out company.

Published September 2017

DOI: 10.3310/hta21530

Plain English summary

Breathing retraining exercises in the management of asthma in adults

Health Technology Assessment 2017; Vol. 21: No. 53

DOI: 10.3310/hta21530

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Most adults with asthma have symptoms that affect their lives, despite taking appropriate medications, and many are interested in non-drug approaches. We compared breathing retraining delivered through 'face-to-face' physiotherapy sessions with both a self-guided programme [digital versatile disc (DVD) plus booklet] of breathing retraining that could be carried out at home and usual care. Adults with incompletely controlled asthma in primary care were randomised into one of these groups. They were assessed before starting the intervention and at 3, 6 and 12 months afterwards for asthma quality of life (QoL) and were also assessed with other questionnaire and physiological measures; in addition, qualitative interviews were carried out to obtain the perspectives of patients and an economic evaluation was performed.

We recruited 655 volunteers from general practice surgeries. QoL significantly improved in both active groups compared with usual care, with equivalent improvements between active groups. Patients reported feeling that the exercises were helpful and acceptable, although some participants who received the DVD would have appreciated talking to a physiotherapist. Lung function and inflammation were unaffected. There were consistent improvement trends in the active arms for symptom scores, rescue medication use, anxiety and depression and asthma attacks, but these improvements were not statistically significant and so could have occurred through chance. Asthma-related costs were lower in both of the active groups.

People with asthma felt and functioned better following breathing retraining delivered by either a DVD or a physiotherapist and programme costs were exceeded by savings from better asthma control, with the DVD being most cost-effective. Patients still had asthma but coped with it better. These simple exercises can be carried out at home and have the potential to improve asthma control and save money.

HTA/HTA TAR

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.236

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the Clarivate Analytics Science Citation Index

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hta

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 09/104/19. The contractual start date was in November 2011. The draft report began editorial review in October 2016 and was accepted for publication in May 2017. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health.

© Queen's Printer and Controller of HMSO 2017. This work was produced by Thomas et al. under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Health Technology Assessment Editor-in-Chief

Professor Hywel Williams Director, HTA Programme, UK and Foundation Professor and Co-Director of the Centre of Evidence-Based Dermatology, University of Nottingham, UK

NIHR Journals Library Editor-in-Chief

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the EME Programme, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

Professor Matthias Beck Chair in Public Sector Management and Subject Leader (Management Group), Queen's University Management School, Queen's University Belfast, UK

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Director, Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk